Dear Editor,

Ureaplasma urealyticum, Ureaplasma parvum, Mycoplasma hominis and Mycoplasma genitalium are potentially pathogenic species and playing an etiologic role in both genital infections and male infertility. Reports are however controversial about the effects of genital mycoplasmas and ureaplasmas infections or colonisation on semen andrology variables.

Please find enclosed our manuscript entitled The correlation of Ureaplasma urealyticum, Ureaplasma parvum, Mycoplasma hominis and Mycoplasma genitalium infections with male infertility.

I believe that my manuscript, dealing with the prevalence of genital bacteria (genital mycoplasmas and genital ureaplasmas) in relation to male infertility, interests the audience of your journal. This manuscript has not and will not be offered elsewhere for possible publication as long as it is under BMC Infectious Diseases consideration. All authors have seen and approved the content and have contributed significantly to the work.

Sincerely yours,

Radhouane GDOURA